Author’s response to reviews

Title: Leisure and religious activity participation and mental health: gender analysis of older adults in Nepal

Authors:

Ramraj Gautam (rr-gautam@umin.ac.jp)
Tami Saito (tami@m.u-tokyo.ac.jp)
Ichiro Kai (ichirokai-tky@umin.ac.jp)

Version: 3 Date: 10 October 2007

Author’s response to reviews:

October 10, 2007
Dr. Annabel Phillips
Senior Assistant Editor
BMC Public Health
The BioMed Central Editorial Team
Middlesex House
34-42 Cleveland Street
London W1T 4 LB

Dear Dr. Phillips,

Thank you for accepting our manuscript MS: 1610376730138895
Leisure and religious activity participation and mental health: gender analysis of older adults in Nepal.

Reviewer: Annie Rochette

Discretionary Revisions (which the author can choose to ignore):
1. On page 7. Objective of the study: please be careful in using the term "impact" on the mental health; as this study does not allow to differentiate if participation impact on mental health or the reverse, mental health may impact on participation...

Author Response: On page 7, "impact" is changed to "relation to". Thank you.

Ramraj Gautam
(on behalf of co-authors)